Author’s response to reviews

Title: Analgesic principle from Curcuma aeruginosa

Authors:

Chowdhury F Hossain (faiz@ewubd.edu)  
Mohammad Al-Amin (2006alamin@gmail.com)  
Abu Sadat M Sayem (sayem062@yahoo.com)  
Ismail H Siragee (ismailsiragee@gmail.com)  
Asif M Tunan (tunan_but@yahoo.com)  
Fahima Hassan (trishi_8a@hotmail.com)  
Shamsun N Khan (nahar305@yahoo.com)  
Edward L Organ (delo@ewubd.edu)  
Gazi Nurun N Sultana (nngazi@gmail.com)

Version: 2  
Date: 20 December 2014

Author’s response to reviews: see over
Dear Editor:

As per the suggestion of your office by an email of December 15, 2014, subject: “MS: 1042109311533296 - revisions required prior to editorial assessment”, I included the Authors' contributions before the Acknowledgements section and also inserted line numbers in the manuscript. Now I am uploading the corrected version herewith.

I would appreciate very much if you kindly consider it for publication.

Thank you.

Sincerely,

Chowdhury Faiz Hossain, Ph.D.
Chairperson and Professor,
Department of Pharmacy
East West University
A/2, Jahurul Islam City, Aftabnagar, Dhaka-1212, Bangladesh.
Phone: 09666775577 ext. 211 and 212

Email: cfhossain@yahoo.com/faiz@ewubd.edu